\begin{table}%t1
\caption{\label{tab2}Radio properties of the core galaxies.}
\par
%\centerline
{
\begin{tabular}{c| c c c c c | c}
\hline\hline
Name& $F_{\rm core}$ & $F_{\rm 74~MHz}$ & $F_{\rm 1.4~GHz}$ & $F_{\rm 5~GHz}$ & $P_{\rm c}$ & radio structure\\
    &  mJy           &  mJy             & mJy               & mJy             &       &                \\ 
\hline
UGC~5902  & 0.73$^{a}$  &        & 2.4   & 0.7   &  0.30   & VLA unresolved$^{e}$ \\          
UGC~7203  & 4.8$^{b}$   &        & 5.6   & 4.5   &  0.86   & VLBI unresolved$^{m}$ \\           
UGC~7629  & 4.8$^{a}$   &        & 256   & 95	 &  0.02   & twin kpc jets$^{j}$\\            
UGC~7760  & 131.7$^{b}$ &        & 100   & 121   &  1.30   & twin pc jets$^{e}$ \\              
UGC~7898  & 12.5$^{a}$  & 449    & 29.1  & 24	 &  0.43   & twin kpc jets$^{k}$ \\           
UGC~8745  & 10.5$^{a}$  & 470    & 78.4  & 20	 &  0.13   & twin kpc jets$^{l}$\\            
UGC~9655  & 1.2$^{a}$   &        & 14.8  & 2.1   &  0.08   & VLA unresolved$^{h}$\\           
\hline   	      	     	     	    	                                                            
UGC~0968  & 1.5$^{c}$   &        & 3.1   & 1.4   &  0.48   & VLBI unresolved$^{i}$ \\           
UGC~6297  & 2.7$^{a}$   &        & 6.9   & 2.6   &  0.39   & VLA unresolved$^{h}$ \\          
UGC~7386  & 159.8$^{b}$ & 343    & 385   & 351   &  0.41   & pc jet$^{f,g}$ \\                  
UGC~7797  & 20.4$^{a}$  &        & 36.8  & 21	 &  0.55   & twin pc jets$^{h}$\\               
UGC~7878  & 5.3$^{a}$   & 692    & 77.8  & 45	 &  0.07   & twin kpc jets$^{k}$\\            
UGC~9706  & 6.4$^{d}$   &        & 21.0  & 5.3   &  0.30   & pc jet$^{i}$ \\                    
UGC~9723  & 12.7$^{b}$  &        & 21.8  & 7.4   &  0.58   & VLBI unresolved$^{n}$\\
\hline
\end{tabular}}
\par
\medskip 
Column description: (1) name; (2) core flux at 5 GHz in mJy. References: (a)
  Baldi et al. 2009, in preparation; (b) \citet{nagar01}; (c) \citet{nagar05}
  (VLBA); (d) \citet{filho04} at 8 GHz; ; (3) total flux at 74 MHz in mJy; (4)
  total flux at 1.4 GHz from \citet{condon02} in mJy; (5) total flux at 5 GHz
  from \citet{wrobel91b} and \citet{wrobel91a} in mJy; (6) core dominance, (7)
  radio morphology.  References: (e) \citet{nagar02}; (f) \citet{jones84}; (g)
  \citet{falcke00}; (h) \citet{nagar05}; (i) \citet{filho04}; (j)
  \citet{ho01c}; (k)~\citet{stanger86}; (l) \citet{hummel84}; (m)
  \citet{anderson04}; (n) \citet{anderson05}.
\end{table}